Author's response to reviews

Title: Multiple right-sided pulmonary nodules. Metastatic cancer or resectable early stage tumor?

Authors:

Ugo Cioffi (ugo.cioffi@policlinico.mi.it)
Federico Raveglia (ravegliafederico@tiscali.it)
Matilde De Simone (matilde.desimone@policlinico.mi.it)
Vincenzo Valenti (uo.pneumologia.psd@grupposandonato.it)
Michele M Ciulla (michele.ciulla@unimi.it)
Alessandro Baisi (alessandro.baisi@unimi.it)

Version: 2 Date: 28 June 2011

Author’s response to reviews:

To the Editor,

thank you for having promptly revised our Case Report.

We upload the revised version according to the suggestions provided.

Rev1, page 4, line 8, the phrase "Improperly, only one nodule underwent needle aspiration and it was targeted to.." has been added.

Rev2, the title has been changed. The term "control" has been changed with "new" in the abstract. Page 6, line 6, the term "invasivity" has been changed with "invasiveness".